Author's response to reviews

Title: Lack of association between gene polymorphisms of Angiotensin converting enzyme, Nod-like receptor 1, Toll-like receptor 4, FAS/FASL and the presence of Helicobacter pylori-induced premalignant gastric lesions and gastric cancer in Caucasians.

Authors:

Juozas Kupcinskas (j_kupcinskas@yahoo.com)
Thomas Wex (Thomas.Wex@med.ovgu.de)
Jan Bornschein (jan.bornschein@med.ovgu.de)
Michael Selgrad (michael.selgrad@med.ovgu.de)
Marcis Leja (cej@latnet.lv)
Elona Juozaityte (elona.juozaityte@kmuk.lt)
Gediminas Kiudelis (gedikiud@takas.lt)
Laimas Jonaitis (laimasji@takas.lt)
Peter Malfertheiner (peter.malfertheiner@medizin.uni-magdeburg.de)

Version: 3 Date: 23 August 2011

Author's response to reviews:

Dear Colleague,

the manuscript was revised according to the last Email from August 22nd.
- tracking/highlights were removed
- introduction was changes into background
- numbering and names of additional files/tables were adjusted and corrected

Best regards,
Thomas Wex